Author’s response to reviews

Title: Management of fever in Australian children: a population-based sample survey

Authors:

Joanna Holt (joanna.holt@mq.edu.au)
Leslie White (leslie.white@health.nsw.gov.au)
Gavin Wheaton (gavin.wheaton@sa.gov.au)
Helena Williams (hwilliams@russellclinic.com.au)
Shefali Jani (Shefali.Jani@health.nsw.gov.au)
Gaston Arnolda (gaston.arnolda@mq.edu.au)
Hsuen Ting (pei.ting@mq.edu.au)
Peter Hibbert (peter.hibbert@mq.edu.au)
Jeffrey Braithwaite (jeffrey.braithwaite@mq.edu.au)

Version: 2 Date: 13 Nov 2019

Author’s response to reviews:

Brock Madsen
Editor
BMC Pediatrics

Dear Brock Madsen,


On behalf of my co-authors I wish to thank you for the opportunity to revise our work for resubmission of ‘Management of fever in Australian children: a population-based sample survey’ for publication in BMC Pediatrics.

The manuscript has been revised in accordance with the reviewers’ comments and the adjustments have been made, and a clean version has been uploaded. I can confirm that the figures meet the required resolution and conforms to the journal style.
We believe this manuscript, if you think it fits with BMC Pediatrics, would make a valuable contribution because of its outstanding importance to pediatric care. We believe that our research findings will be of interest to readers of BMC Pediatrics, including clinicians who treat children with fever in the UK and internationally. The manuscript comprises original, unpublished material which is not under consideration elsewhere. We have no conflict of interest in connection with this manuscript.

Yours sincerely

Jeffrey Braithwaite, BA, DipLR, MBA, MIR (Hons), PhD, FIML, FCHSM, FFPHRCP (UK), FAcSS (UK), Hon FRACMA, FAHMS
Professor and Founding Director
Australian Institute of Health Innovation